Author’s response to reviews

Title: Factors associated with sterilization use among women leaving a U.S. jail: A mixed methods study

Authors:

Megha Ramaswamy (mramaswamy@kumc.edu)
Patricia J Kelly (kellypj@umkc.edu)

Version: 2 Date: 30 October 2013

Author’s response to reviews:

Dear Editors, the only change we made to the revised manuscript is the inclusion of a Competing Interests section. Thanks,

Megha